
-----------------------------------------------------------------------------
          Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
       Mon Aug 27 17:14:41 GMT 2018
Year: 18  Doy: 239
Observer: berkey
WEATHER COMMENT: berkey: Mon Aug 27 17:17:17 GMT 2018
Temp: 45.1f, Humidity: 94%, Pressure: 28.678in, Wind: 0mph from 126degs, Skies: patchy cirrus.  We may get some data today
. (but maybe not).
___end___
GENERAL COMMENT BY berkey: Mon Aug 27 17:29:55 GMT 2018
PM Washed Kcor O1
___end___
Mon Aug 27 17:30:58 GMT 2018 SGS Alignment complete
Mon Aug 27 17:50:43 GMT 2018 Kcor Focus/alignment program exited
GENERAL COMMENT BY berkey: Mon Aug 27 17:54:21 GMT 2018
Windows open upstairs.  However with 0 wind this may be less effective than normal at removing heat from the dome.
___end___
GONG COMMENT BY berkey: Mon Aug 27 18:01:24 GMT 2018
Cover removed from Gong turret
___end___
GENERAL COMMENT BY berkey: Mon Aug 27 18:08:38 GMT 2018
Dome closed.
___end___
GENERAL COMMENT BY berkey: Mon Aug 27 20:20:05 GMT 2018
Cosmic gps antenna back out in the lava flow.
___end___
GENERAL COMMENT BY berkey: Mon Aug 27 21:52:43 GMT 2018
Heavy down pour.
___end___
GENERAL COMMENT BY berkey: Mon Aug 27 22:28:21 GMT 2018
Fire extinguisher inspection today.
___end___
GENERAL COMMENT BY berkey: Tue Aug 28 01:28:49 GMT 2018
Got 3 ok kcor images this morning :) Neat data again.

T-storms on the west side are starting to close roads.  Time to go or maybe slightly past time.
___end___
ONSITE STAFF:  berkey


